vc/5-Nov-03
Week 45 starting 03 nov 03

SM18 Operation SCHEDULE Meeting (Tuesdays)

Schedule for this week : 
	BENCH
	CURRENT

MAGNET
	STATUS 

Today(04/11/03)
	Forward look,   & Next Magnet 

	TBA1
	NOT Active
	
	Bench de-activated – preparation for D bench

“Park” here 2001 ( moved from D bench )

	TBA2
	3055
	Start to connect
	Magnet 3055 should or just arrived

	TBC1
	1048
	Under connection
	

	TBC2
	SSS3
	To connect
	

	TBE1
	2026
	Cooling after TH.Cycle
	COLD Tests AFTER THERMAL CYCLE foresee on Wednesday

	TBE2
	3008
	Cold 
	Continue cold tests

	TBF1
	1052
	Warm; to remove
	1021 next ?

	TBF2
	
	CFB under repair ?
	


Test Coordinator this wek:  V.Chohan;  MTM  Tech. on call:  (16-3602) : 
· Situation From Cryogenics:  J-P Lamboy or ACR Rep.

· Other General Issues
1. Resistance M’ments & Traveller

2. new todo list  numbers…
3. AOB

Current (wed 5thNov  pm) Bench Status:

	SM18 LHC Magnet Test Facility [image: image1]


	TBA1
	_
	_
	 


	TBA2
	MBAR3055
	Prep7 HV Initial @ Warm
	300 K


	TBB1
	_
	_
	 


	TBB2
	_
	_
	 


	TBC1
	MBBL1048
	Prep12 Pumping + Cool Down
	Cooling


	TBC2
	SSS003
	Prep2 HV Test Warm
	300 K


	TBD1
	_
	_
	 


	TBD2
	_
	_
	 


	TBE1
	MBAR2026
	PT6.1 Training 1
	1.9 K


	TBE2
	MBAR3008
	MM13 TRU Disconnect
	Warming


	TBF1
	MBAR1052
	ICS 3Traveller Last
	300 K


	TBF2
	MBBR1039
	Prep3 HV Initial @ Warm
	300 K
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Problems of Last Week

	Date
	Problem
	Type
	Application/soft.
	Action/advise

	Tuesday 28_M


	TBF2:  LHC cycle : motor giving problem

CPU error on MMP
	MTM/CO
	
	LHC cycle aborted and repeated later

Problem solved by Pedro?.

	Tuesday 28_M


	TBE1: HV test at warm failed on dipole-QH-ground.
	MTM
	
	Problem with ICS rack?

	Tuesday 28_A


	TBF2: After a Min energy quench alarm in AQA : High DT (144 V) on D2_L.

Voltage sum not respected on D2 pole
	MTM
	
	Stephane changed the card.

	Tuesday 28_N


	Cryo problems : many stops during the nights
	ACR
	
	

	Tuesday 28_N

Wednesday 29_M 
	TBF2: Motor problem son TRU when launching the auto-test. Encoders errors, motor busy
	
	
	Origin on investigation. MMP launched by switching off/on the power supply of the crate

	Wednesday 29_M 
	TBF1: IAP @warm all LF heaters.

Wrong storage of the file
	MTM
	IAP
	The correct active bench has to be selected. Active bench selected automatically when the sequence is selected?

	Wednesday 29_A 
	TBC1: diode test reverse mode,

Impossible to transfer the file on sun MTA19
	MTM
	
	O.Berrig modified badly the .tchrc

	Wednesday 29_N 
	TBA2: MM0: MUX problem during auto test
	MTM
	
	

	Thursday 30_M 
	TBA2: during pre-cycle (LHC) power application was hung and not communication to power converter.
	AB/PO
	Power
	LHC cycle re-started

	Thursday 30_N 
	TBA2: de-excitation test: After 9000 A the current did not raise further.
	AB/PO
	Power
	The button execute ramp has to be clicked.

	Friday 31_N 
	TBE1: IAP at cold : offset error in Vgas_inT_lead, D1_L_U, D1-D2. 8 errors in QA compensation.
	MTM
	IAP
	Adjusted .

	Friday 31_N 
	TBE1: Provoked quench at 1.5 kA (HF). Quench in Vbus ext (repeated 3 times).
	
	
	Symmetrical quench?

	Saturday 1_M
	TBE1: Min energy at 3 kA (diode test)

Data for diode test not in order ???
	MTM
	
	

	Saturday 1_A
	TBF1: T coil .D2S01_C opened, (c coil in shaft 25 broken), wrong value in D1_S07
	MTM
	
	

	Saturday 1_A
	TBF1: HV test cold before test: Cryo _X1 not OK

Communication problem between PLC  cryo and PLC
	ACR/MTM
	
	

	Saturday 1_N
	Cryo problem: Difficulty to control the pressure of 3 magnets due to lack of He pressure
	ACR
	
	

	Saturday 1_N

Sunday_A
	TBF1: Quench at 1.5 kA and 3 kA: Signals in D2 are zero and D1-D2 negative!

Non consistence between the signals
	MTM
	
	Test stopped.

MBAL template used instead of MBBR

	Saturday 1_N
	TBE1: LHC cycle aborted because the cable were scratching with the cover.
	MTM
	
	Test stopped.

	Sunday 2_A
	TBA2: IAP final at warm  power supply is tripping

No automatic trigger 
	MTM
	IAP
	Test stopped.

	Sunday 2_A
	TBA2: QH resistance : values  are found to be too high.

Problem in the connectors of the measurement box 

Request of cables fitted to measured the cryoheater (as for the temperature sensor).

One connector in the cable of the 4-wires measurement for the resistance is out of work.


	MTM
	IAP
	Box replaced. Measurement performed again.


